Author's response to reviews

Title: Prevalence of diphtheria toxoid IgG antibodies in children, adolescents and adults in Poland

Authors:

Aleksandra A Zasada (azasada@pzh.gov.pl)
Waldemar Rastawicki (wrastawicki@pzh.gov.pl)
Natalia Rokosz (nrokosz@pzh.gov.pl)
Marek Jagielski (mjagielski@pzh.gov.pl)

Version: 4 Date: 14 June 2013

Author's response to reviews: see over
Author's response to reviews

Title: Prevalence of diphtheria toxoid IgG antibodies in children, adolescents and adults in Poland

Authors:

Aleksandra A Zasada (azasada@pzh.gov.pl)
Waldemar Rastawicki (wrastawicki@pzh.gov.pl)
Natalia Rokosz (nrokosz@pzh.gov.pl)
Marek Jagielski (mjagielski@pzh.gov.pl)

Version: 2 Date: 23 May 2013

Author's response to reviews:

Dear Dr. Sheryl Ramos,

I am submitting the revised version of the manuscript „Prevalence of diphtheria toxoid IgG antibodies in children, adolescents and adults in Poland” (MS: 8640087749941999).

Information concerning the approval of the studies given by Bioethics Committee as well as information concerning written informed consent given by participants have been added in the Methods section.

Sincerely,

on behalf of all authors,

Aleksandra Zasada